Author’s response to reviews

Title: Comparative study of sickle cell anemia and hemoglobin SC disease: clinical characterization, laboratory biomarkers and genetic profiles

Authors:

Milena Aleluia (milena.aleluia@gmail.com)
Teresa Fonseca (ruiteresafonseca@uol.com.br)
Regiana Souza (regianaquinto@hotmail.com)
Fábia Neves (fabia_neves83@yahoo.com.br)
Caroline Guarda (cguarda4@hotmail.com)
Rayra Santiago (rayrasantiago@hotmail.com)
Bruna Cunha (brunal.a.cunha@gmail.com)
Camylla Figueiredo (myllavilas@hotmail.com)
Sânzio Santana (sanziobiomedico@hotmail.com)
Silvana Paz (spaz@bahia.fiocruz.br)
Júnia Ferreira (juniadutra@gmail.com)
Bruno Cerqueira (brunoveloso2@hotmail.com)
Marilda Gonçalves (mari@bahia.fiocruz.br)

Version: 2 Date: 11 Apr 2017

Author’s response to reviews:

April, 11th, 2017

To: Nawsheen Boodhun, MD
BMC Hematology Editorial Office
bmchematol@biomedcentral.com
From: Authorship of the manuscript entitled “Comparative study of sickle cell anemia and hemoglobin SC disease: clinical characterization, laboratory biomarkers and genetic profiles” (BMC Hematology - BHEM-D-16-00045R1).

Subject: Rebuttal letter

Dear Dr. Boodhun,

Thank you for allowing us to revise our manuscript entitled “Comparative study of sickle cell anemia and hemoglobin SC disease: clinical characterization, laboratory biomarkers and genetic profiles” (BMC Hematology - BHEM-D-16-00045R1). We thank the reviewer for the careful perusal of our manuscript and thoughtful comments, which certainly have improved it. We would like to say that we accepted all comments raised by the review. We have addressed each reviewer comments, as described below. We would like to thank you for your kindly to let us revised our previous version. We remain available for further clarifications and adjustments that are necessary and we hope that the manuscript is now suitable for publication.

Sincerely,

Marilda de Souza Goncalves, Ph.D.